Author's response to reviews

Title: Knowledge and future preference of Chinese women in a major public hospital in Hong Kong after undergoing non-invasive prenatal testing for positive aneuploidy screening: a questionnaire survey

Authors:

Kam On KOU (kouko@ha.org.hk)
Chung Fan POON (pcf301@ha.org.hk)
Wai Ching TSE (twc320@pmh.ha.org.hk)
Sui Lam MAK (msl061@ha.org.hk)
Kwok Yin LEUNG (leungky1@ha.org.hk)

Version: 5 Date: 31 July 2015

Author's response to reviews: see over
Cover letter for revisions:

Mr Jason Pepito
Editor
BioMed Central Editorial

Dear Mr. Jason Pepito

MS: 1294366479159195 - Knowledge and future preference of Chinese women in a major public hospital in Hong Kong after undergoing non-invasive prenatal testing for positive aneuploidy screening: a questionnaire survey

Thank you very much for processing our manuscript and the valuable comments from the editors. These will help improve the quality of our manuscript. We have revised our manuscript according to the comments and highlighted the changes.

1. We have deleted the short title from the title page.

2. We have deleted the section “Disclosure” from the manuscript. Ethical approval was detailed in the Methods section with full name of the IRB provided.
   ‘The study protocol was approved by the Research Ethics Committee (Kowloon Central / Kowloon East), Hospital Authority, Hong Kong. This research project conforms to the provisions of the Declaration of Helsinki.’

3. We have added the section “Competing interests”. (lines 306-310).
   Competing interests
   There is no financial support or relationships that may pose a potential conflict of interest, and it is confirmed that the results of this manuscript have not been distorted by research funding or conflicts of interest. The authors declare that they have no competing interests.

4. We confirmed that informed consent to participate in the study was obtained from all participants. (lines 91-94).
   Informed consent was obtained from every participant in this study.

5. The questionnaire is attached as an additional file.

We hope that our manuscript is now acceptable for publication in your journal.
Regards,

KY
Dr. KY Leung
Corresponding author